Author's response to reviews

Title: Lipoarabinomannan in urine during tuberculosis treatment: association with host and pathogen factors and mycobacteriuria

Authors:

Robin Wood (Robin.Wood@hiv-research.org.za)
Kimberly Racow (Kimberly.Racow@hiv-research.org.za)
Linda-Gail Bekker (Linda-Gail.Bekker@hiv-research.org.za)
Keren Middelkoop (Keren.Middelkoop@hiv-research.org.za)
Monica Vogt (Monica.Vogt@hiv-research.org.za)
Barry N Kreiswirth (Kreiswba@umdnj.edu)
Stephen D Lawn (Stevelawn@yahoo.co.uk)

Version: 4 Date: 14 February 2012

Author's response to reviews:

Dear Editors of the BMC Infectious Diseases journal,

Thank you for providing us with the final formatted version of our manuscript; however, we have a few minor edits we would like to submit.

-Page 4-"The aforementioned analyses were performed in accredited laboratories of the South African National Health Laboratory Service (NHLS). Sputum samples collected for direct sputum smears during the course of routine management were additionally requested to be cultured for Mtb at an accredited TB laboratory of the NHLS."

-Page 7, Table 1-Indented "Pulmonary and Extra-Pulmonary" and changed the "a" to superscripts after "Median CD4 cells/µL. [IQR]" and "ART at start of TB treatment"

Thank you,
Kimberly Racow